

# G OPEN ACCESS

**Citation:** Kroidl I, Chachage M, Mnkai J, Nsojo A, Berninghoff M, Verweij JJ, et al. (2019) *Wuchereria bancrofti* infection is linked to systemic activation of CD4 and CD8 T cells. PLoS Negl Trop Dis 13(8): e0007623. https://doi.org/10.1371/journal. pntd.0007623

Editor: Subash Babu, NIH-NIRT-ICER, INDIA

Received: January 30, 2019

Accepted: July 10, 2019

Published: August 19, 2019

**Copyright:** © 2019 Kroidl et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

**Funding:** The EMINI study was funded by the European Union (SANTE/2004/078-545 and SANTE/2006/129-931). The WHIS study received funding by the German Research Foundation (DFG grant WSA 1878/1-1) with additional support by the European Community's Seventh Framework Programme (FP7/2007-2013 and FP7/2007-2011 und EC-GA no. 241642. The testing for W. bancrofti was supported by a grant from the RESEARCH ARTICLE

# *Wuchereria bancrofti* infection is linked to systemic activation of CD4 and CD8 T cells

Inge Kroidl<sup>1,2,3</sup>\*, Mkunde Chachage<sup>1,2</sup>, Jonathan Mnkai<sup>2</sup>, Anthony Nsojo<sup>2</sup>, Myrna Berninghoff<sup>1</sup>, Jaco J. Verweij<sup>4</sup>, Lucas Maganga<sup>2</sup>, Nyanda E. Ntinginya<sup>2</sup>, Leonard Maboko<sup>2</sup>, Petra Clowes<sup>1,2</sup>, Michael Hoelscher<sup>1,2,3</sup>, Elmar Saathoff<sup>1,3</sup>, Christof Geldmacher<sup>1,3</sup>

Division of Infectious Diseases and Tropical Medicine, University Hospital, LMU Munich Germany,
National Institute for Medical Research (NIMR)-Mbeya Medical Research Center (MMRC), Mbeya,
Tanzania, 3 German Center for Infection Research (DZIF), partner site Munich, Germany, 4 Laboratory for
Medical Microbiology and Immunology, Elisabeth Tweesteden Hospital, Tilburg, The Netherlands

So These authors contributed equally to this work.

\* ikroidl@Irz.uni-muenchen.de

# Abstract

## Background

Susceptibility to HIV has been linked to systemic CD4+ T cell activation in cohorts of seronegative individuals with high HIV-exposure risk. We recently described an increased risk of HIV transmission in individuals infected with *Wuchereria bancrofti*, the causative agent for lymphatic filariasis, in a prospective cohort study. However, the reason for this phenomenon needs further investigation.

### Methodology/Principal findings

Two-hundred and thirty-five HIV negative adults were tested using Trop Bio ELISA for detection of *W. bancrofti* infection and Kato Katz urine filtration and stool based RT-PCR for detection of soil transmitted helminths and schistosomiasis. FACS analysis of the fresh peripheral whole blood was used to measure T cell activation markers (HLA-DR, CD38), differentiation markers (CD45, CD27), markers for regulatory T cells (FoxP3, CD25) and the HIV entry receptor CCR5. Frequencies of activated HLA-DR<sup>pos</sup> CD4 T cells were significantly increased in subjects with *W. bancrofti* infection (n = 33 median: 10.71%) compared to subjects without any helminth infection (n = 42, median 6.97%, p = 0.011) or those with other helminths (*Schistosoma haematobium, S. mansoni, Trichuris trichiura, Ascaris lumbricoides*, hookworm) (n = 151, median 7.38%, p = 0.009). Similarly, a significant increase in HLA-DR<sup>pos</sup>CD38<sup>pos</sup> CD4 T cells and effector memory cells CD4 T cells (CD45RO<sup>pos</sup>CD27-<sup>neg</sup>) was observed in filarial infected participants. Multivariable analyses further confirmed a link between *W. bancrofti* infection and systemic activation of CD4 T cells independent of age, fever, gender or other helminth infections.

### **Conclusions/Significance**

*W. bancrofti* infection is linked to systemic CD4 T cell activation, which may contribute to the increased susceptibility of *W. bancrofti* infected individuals to HIV infection.

German Ministry of Science (BMBF, grant number 01KA0904, Surveillance of lymphatic filariasis (SOLF)). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Author summary

The importance of CD4 T cell activation for HIV susceptibility has been emphasized in several studies focusing on HIV transmission and prevention. Particularly, activated HLA-DR+ CD4 T cells may play a major role in HIV susceptibility. In this analysis we describe systemic activation of CD4 T cells in individuals infected with *W. bancrofti* the causative agent of lymphatic filariasis. This helminth disease leads to debilitating pathology in some of the individuals; however, the majority of infected persons remain asymptomatic. We recently described an increased HIV incidence in subjects infected with *W. bancrofti* compared to uninfected individuals from the same area. To decipher underlying reasons for this phenomenon, we measured immune activation parameters in CD4 and CD8 T cells. The increased percentage of HLADR positive and HLADR/CD38 positive CD4 T cells and also effector memory CD4 T cells that we describe here could be a possible mechanism to explain our previous findings of increased HIV incidence in individuals infected with this filarial nematode.

#### Introduction

The human immunodeficiency virus (HIV) epidemic and high HIV transmission rates continue to affect large parts of the world [1]. The disproportionately high prevalence of HIV in communities of sub-Saharan Africa has led to the hypothesis that concomitant helminth infections augment the risk of HIV transmission by increasing systemic immune activation [2–5]. Helminths are parasitic worms that primarily affect the world's poor and cause chronic infections in one-fourth of the world's population [6]. Different helminth infections can induce a diverse array of clinical symptoms and pathology. In addition to their associated morbidity, several distinct immunological changes have been described for different helminth species [7], some of which might potentially increase HIV susceptibility.

Recently, we demonstrated for the first time an increased risk of HIV acquisition in individuals infected with the filarial nematode *Wuchereria bancrofti*, the agent that causes lymphatic filariasis [8]. Our data shows that *W. bancrofti* is an independent risk factor for HIV acquisition even when adjusting for sexual behavior, condom use, circumcision, marital status, age and gender, among other factors.

Infections with *W. bancrofti* are chronic and persist over many years despite antihelminthic treatment. The majority of infected individuals remain asymptomatic; however, 10 to 30% develop filarial pathology with lymphedema or hydrocele. Different clinical outcomes are associated with distinct immunological phenotypes. The early phase of a filarial infection is characterized by proinflammatory cytokines [9], followed by anti-inflammatory cytokines such as IL-10 and TGF- beta during the chronic phase of infection in clinically asymptomatic individuals [10, 11]. In patients with chronic filarial pathology, a lack of regulatory T cells and increased Th1 and Th17 proinflammatory responses were documented [12]. Members of the family of vascular endothelial growth factors, as well as IL-1beta, TNF-alpha, and IL-6 have been implicated in lymphangiogenesis [13–15]. Individuals with lymphedema showed increased frequencies of HLA-DR<sup>pos</sup> CD8 T cells [16]; however no differences in CD4 T cell maturation or expression of HLADR have yet been described for *W. bancrofti*-infected individuals.

Activated CD4 T cells are a major cellular reservoir for continuous HIV replication *in vivo*, as shown in several studies [17–20]. In order to address the hypothesis that the increased HIV transmission risk in *W. bancrofti* -infected subjects is linked to increased systemic CD4 T cell

activation, we studied expression of activation (HLA-DR, CD38) and maturation (CD27, CD45RO) markers in CD4 and CD8 T cell populations, as well as T regulatory CD4 T cells and HIV entry receptor (CCR5) in relation to *W. bancrofti* infection status.

# Material and methods

#### Ethics

This study was approved by the ethics committees of the Tanzanian National Institute for Medical Research, Mbeya Medical Research and Ethics Committee and Munich University Hospital, and was conducted according to the principles expressed in the Declaration of Helsinki. All participants recruited in the study were adults above 18 years of age who provided written informed consent before enrolment into the study.

#### Study volunteers

A total of 386 adult study participants from the "Evaluating and Monitoring the Impact of New Interventions" (EMINI) general population cohort from the Mbeya region in Southwest Tanzania were enrolled into the prospective Worm-HIV-Interaction-Study (WHIS) based on their helminth and HIV infection status between October 2009 and March 2012 [21]. After excluding 136 HIV-infected study participants, 2 individuals without filariasis results, and 13 participants with invalid flow cytometry data, results from 235 HIV negative volunteers (137 women and 98 men) were analyzed. In the study area, governmental treatment of *W. bancrofti* started in 2009. However, none of the 235 participants reported participation in the mass drug administration program.

### Methods

This study uses the blood, urine and stool specimens that were collected from each participant at baseline. HIV status was determined using HIV 1/2 STAT-PAK (Chem-Bio Diagnostics Systems) and positive results were confirmed using ELISA (Bio-Rad) with discrepancies being resolved by Western Blot (MPD HIV Blot 2.2, MP Biomedicals). Forty ml of venous blood were drawn from each participant using anticoagulant tubes (CPDA, EDTA; BD Vacutainer). Fresh blood samples were processed within 6 hours of the blood draw at the NIMR-MMRC laboratories.

**Diagnosis of helminth infections.** *W. bancrofti* infection was diagnosed using an ELISA specific for circulating filarial antigen (TropBio Og4C3 serum ELISA, Townsville, Australia) as described previously [22] in bio banked cryopreserved plasma, information on the presence of microfilariae was not available. Four individuals showed *W. bancrofti*-associated pathology: High levels of circulating filarial antigen were measured in two individuals with hydrocele, whereas another two individuals with lymphedema of the legs had a negative ELISA result. As this phenomenon is common in filarial associated lymphedema, both participants were grouped as *W. bancrofti* infected. *Schistosoma haematobium* infection was diagnosed by microscopic examination of a filtered urine sample (20 ml). Fresh stool specimens were used for Kato-Katz diagnosis of intestinal nematode and *S. mansoni* infections [23–25]. Briefly, two Kato-Katz thick smears (41.7 mg each) were prepared from each fresh stool. Kato-Katz slides were microscopically examined for helminth eggs by experienced technicians within one hour (for hookworm eggs) and within two days (for other helminth eggs) after slide preparation. Helminth infection was defined as the presence of at least one worm egg in the examined samples.

In addition, a multiplexed RT-PCR was performed for all subjects on fecal DNA samples for simultaneous detection of Schistosoma species, hookworm, Ascaris lumbricoides and Strongyloides stercoralis [26]. For DNA amplification, 5 µL DNA extracted from 0.1 g stool specimens were used as a template in a final volume of 25 µL with PCR buffer (HotstarTaq Master Mix [5 mM MgCl<sub>2</sub> and 2.5 mg bovine serum albumin]; Roche Diagnostics, Almere, The Netherlands), 2 pmol of each A. lumbricoides-specific primer, 5 pmol of each N. americanus-specific primer, 5 pmol of each Schistosoma-specific primer and 2.5 pmol of each S. stercoralis- specific primer. For Necator americanus, A. lumbricoides, S. stercoralis, and Schistosoma-specific double-labelled probes, 1.25 pmol were used. PCR amplification consisted of 15 minutes at 95°C followed by 50 cycles of 15 seconds at 95°C, 30 seconds at 60°C, and 30 seconds at 72°C. Amplification, detection and data analysis were performed at NIMR-MMRC in Tanzania with the Corbett Rotor-Gene 6000 Real-Time PCR System and Corbett Rotor-Gene 6000 Application Software, version 1.7. Negative and positive external control samples were included in each amplification run. The details of all primers and detection probes used in our study are described elsewhere [26-28]. Before testing of actual WHIS stool samples, molecular diagnosis of helminths at NIMR-MMRC in Tanzania was validated externally at the University of Leiden, Leiden, Netherlands, by comparing results for 19 Tanzanian fecal DNA samples. Concordance between the two laboratories was very high: 89% for Schistosoma species (11 of 13 positive samples) and 100% concordance for N. americanus (6 positive samples), A. lumbricoides (3 positive samples), S. stercoralis (2 positive samples) and Ancylostoma duodenale (no positive sample).

**Quantification of eosinophil counts.** An automated complete blood count machine (Beckman Coulter) was used for counting eosinophils. If eosinophil counts were out of range  $(>1000 \times 10^3/\mu l)$ , determination was performed using the differential blood count.

Characterization of markers on CD4 and CD8 T cells in fresh whole blood. Frequencies of activation (HLA-DR, CD38) and maturation (CD27, CD45RO) markers on CD4 and CD8 T cells were determined as previously described [21] in fresh, anti-coagulated whole blood. CD4<sup>pos</sup> T lymphocytes were classified as "naïve" (CD27<sup>pos</sup>CD45RO<sup>neg</sup>), "central memory-like" (CD27<sup>pos</sup>CD45RO<sup>pos</sup>) and effector memory (CD27<sup>neg</sup>CD45RO<sup>pos</sup>) CD4 T cells. Gating strategies are given in the supplement. Blood samples were incubated for 10 minutes with CCR5 PECy7 (natutec, Frankfurt, Germany) followed by 30 minutes incubation using the following fluorochrome labeled monoclonal antibodies for cell surface staining (mABs): CD3 Pac Blue (Becton Dickinson; New Jersey, USA), CD4 Per-CP Cy5.5 (eBioscience, San Diego USA), CD8 V500 or CD8 Amcyan, CD25 phycoerythrin (PE)-Cy7 (eBioscience), CD27 APC-H7, CD45RO APC, HLA-DR FITC and CD38 PE (all from BD). Intracellular forkhead-box-protein P3 (FoxP3) was detected using FoxP3-Alexa Fluor 647 (ebioscience). Stained cells were then finally fixed with 2% paraformaldehyde prior to acquisition. Acquisition was performed on a FACS CANTO II (BD). Compensation was conducted with antibody capture beads (BD) stained separately with the individual antibodies used in the test samples. Flow cytometry data was analyzed using FlowJo (version 9.5.3; Tree Star Inc.).

**Statistical analysis.** GraphPad prism Version 6 was used to graphically compare single parameters between groups. Since most outcomes were not normally distributed, we used the Kruskall Wallis test to compare groups overall and the Mann-Whitney test for comparisons between two groups. To assess the influence of filariasis infection on different markers of immune status, we adjusted for potential confounders using Stata version 14.1 statistics software. We constructed mixed linear regression models with area of residence as a random effect for both uni-variable analysis and multi-variable analysis and adjusted for potential confounders (age, gender, infections with different helminths and the presence of fever in the last 24 hours).





#### Results

Among the 235 HIV negative study subjects, 42 (17.9%) were helminth negative and 193 (82.1%) were helminth positive (Fig 1). *W. bancrofti* infection was detected in 33 subjects (16.1%). Other helminth infections included *S. mansoni* (82 cases, 42.5%), *S. haematobium* (18 cases, 9.3%), *N. americanus* (hookworm) (82 cases, 42.5%), *A. lumbricoides* (52 cases, 26.9%) and *Trichuris trichiura* (38 cases, 19.7%) (Fig 1). Of the 193 individuals with diagnosed helminth infection, 104 (53.9%) were infected with one helminth species, 66 (34.2%) with two species, and 23 (11.9%) with 3 or more helminth species.

Almost all W. bancrofti-infected subjects (32 of 33) were also infected with other helminths (Table 1) impeding an evaluation of only monoinfected individuals. A mean of 2.3 helminths species were detected in W. bancrofti-infected individuals. The composition of the "additional" species in W. bancrofti-infected individuals was almost identical to that in the group harboring other helminth species (Table 1). Therefore we compared the effect of W. bancrofti on T cell activation not only with the group of helminth free individuals, but in addition with the "background" infection of the combined helminth group to distinguish the effect of W. bancrofti from an overall helminth influence on immune activation. The median age in the three groups ranged from 31 to 37 years and the percentage of males was between 39% and 48%. All three groups included less than 10% of individuals with self-reported fever during the last 24 hours (Table 1). As expected, there was a difference in eosinophil count between helminth free individuals and participants with one of the diagnosed helminth species. We found a median number of  $130 \times 10^3$ /µl eosinophils in helminth free individuals compared to  $400 \times 10^3$ /µl in W. *bancrofti*-infected individuals and  $260 \times 10^3$ /µl in participants harboring other helminths. We also found that eosinophil counts increased in parallel with the number of helminth infections (Fig 2).

Table 1. Participant characteristics by helminth status. The group of *W. bancrofti* infected individuals was compared to helminth-free individuals and a group of participant infected with other helminth species. For each group the median age, gender distribution, pathology, the mean number of different helminth species, and the percentage of each type of helminth in the respective group, median eosinophil count and percentage of individuals with fever is given.

	Helminth negative individuals	W. bancrofti—infected individuals	Combined group of individuals infected with other helminths
N	42	33	160
Age median	37.2	31.6	31.4
(range)	(18.4 to 76.0)	(19.1 to 54.0)	(18.0 to 62.4)
Male gender, <u>N (%)</u>	19 (45%)	16 (48%)	63 (39%)
Filarial pathology			
Hydrocele	0	2 (6%)	0
Lymphedema	0	2 (6%)	0
mean number of different helminth species	0	2.3	1.4
Additional helminths			
None	42 (100%)	1 (3%)	0
Schistosoma spec.	0	14 (42%)	82 (51%)
Hookworm	0	12 (36%)	70 (44%)
A. lumbricoides	0	9 (27%)	43 (27%)
T. trichiura	0	13 (39%)	25 (16%)
Eosinophil count *10 <sup>3</sup> /µl			
median	130	400	260
(IQR)	(80 to 230)	(260 to 490)	(150 to 455)
Fever, N (%)	4 (9.5%)	2 (6.1%)	14 (8.8%)

https://doi.org/10.1371/journal.pntd.0007623.t001

#### Increased systemic T cell activation of individuals infected with W. bancrofti

Fresh whole blood was used for flow cytometric assessment of activation and maturation markers on T cells. Frequencies of HLA-DR<sup>pos</sup> CD4 T cells were significantly increased in subjects with *W. bancrofti* infection (n = 33 median: 10.71%) compared to subjects either without any helminth infection (n = 42, median 6.97%, p = 0.016 Mann-Whitney test, univariable analysis, Fig 3A) or the combined group of individuals harboring other helminths (*S. haematobium, S. mansoni, T. trichiura, A. lumbricoides*, hookworms) (n = 151 median 7.38%, p = 0.009). Polyparasitism was not significantly associated with increased frequencies of HLA-DR<sup>pos</sup> CD4 T cells (Fig 3B). *W. bancrofti* infection was associated with increased frequencies of n = 41median 2.84% versus 2.16%, p = 0.035, Fig 4A). Again, polyparasitism was not significantly associated with increased sport for the subjects (Fig 4B).

To adjust the above associations of filariasis infection with immune activation for potentially confounding factors, we estimated multivariable mixed effects models including age, gender, fever and the other helminth infections (*T. trichiura, A. lumbricoides, S. mansoni, S. haematobium* and hookworm) as fixed effects and study site as a random effect in linear regression models. Univariable analysis had demonstrated an influence of these factors on our outcome parameters (Tables 2 and <u>S1–S8</u>). The significant association of *W. bancrofti* infection with HLA-DR<sup>pos</sup> CD4 T cell counts was confirmed in the multivariable analysis (<u>Table 2</u>).

<u>Table 2</u> shows HLA-DR<sup>pos</sup> CD4 T cell counts of all tested helminths. Some other helminth species, for example *A. lumbricoides and T. trichiura also* show elevated HLA-DR<sup>pos</sup> CD4 T cells in uni- and multivariable analysis. This was described and discussed in a previous manuscript about this cohort [21] and is not repeated in this manuscript. In contrast, other helminth species, for example hookworms seem to have no impact on immune activation marker.

Similar uni- and multivariable analyses were performed for HLA-DR<sup>pos</sup> CD8 T cell counts as well as for HLA-DR<sup>pos</sup> CD38<sup>pos</sup> CD4 and HLA-DR<sup>pos</sup> CD38<sup>pos</sup> CD8 T cell counts (Tables 3 and S1, S2, S3).



**Fig 2. Eosinophil counts and polyparasitism.** Eosinophil counts were available for 37 study participants without helminth infection, 87 with one helminth species, 46 with two species, and 7 with three or more helminth species. Increasing numbers of eosinophils were seen with polyparasitism. *Kruskal-Wallis* testing showed significant differences between groups overall (p = 0.0001) as well as *Mann-Whitney* testing to compare the groups separately.

The significant increase in activated T cells in *W. bancrofti*-infected individuals was confirmed for the HLA-DR<sup>pos</sup> CD38<sup>pos</sup> CD4 and CD8 T cells. Frequencies of HLA-DR<sup>pos</sup> CD8 T cell were increased in subjects with *W. bancrofti* (mean 29.5%) infection compared to subjects without filarial infection (mean 20.8%). However, this difference was not significant after adjusting for study site and other confounding factors in the multivariable analysis (Tables 3



**Fig 3.** A: **Expression of HLADR on CD4 T cells in relation to the helminth infection status.** The percentage of HLADR expressing CD4 T cells is shown for the three categories of participants. *Kruskal-Wallis* testing showed significant difference between groups overall (p = 0.0168). *W. bancrofti*-infected individuals had significantly more HLADR expressing CD4 T cells compared to helminth free individuals or study participants infected with other helminths as shown by *Mann-Whitney* test. **B: Association of polyparasitism with activation status.** Number of eosinophils is shown for participants with increasing numbers of helminth species. *Kruskal-Wallis* testing showed no significant difference between groups (p = 0.917) as well as *Mann-Whitney* test.

https://doi.org/10.1371/journal.pntd.0007623.g003



**Fig 4.** A: **Association of helminth status with the frequency of HLA-DR**<sup>pos</sup> **CD38**<sup>pos</sup> **CD4 T cells.** The percentage of HLA-DR<sup>+</sup> CD38<sup>+</sup> expressing CD4 T cells is shown for the three categories of participants. *Kruskal-Wallis* testing showed significant difference between groups overall (p = 0.044). A significant difference was seen between the subgroup of individuals without helminths and the *W. bancrofti*- infected participants (using *Mann-Whitney* testing). **B: Association of polyparasitism with the frequency of HLA-DR<sup>+</sup> CD38<sup>+</sup> CD4 T cells**. The percentage of HLA-DR<sup>+</sup> CD38<sup>+</sup> expressing CD4 T cells is shown for participants with increasing numbers of helminth species. *Kruskal-Wallis* testing showed no significant difference between groups (p = 0.262), as did the Mann-Whitney.

and S2). To give an overview of the different parameters associated with immune activation, which are influenced by infection with *W. bancrofti*, we summarized the results in Table 3 whereas a complete multivariable analysis showing the influence of other factors are given in the supplementary Tables S1–S8.

Apart from T cell activation, we also studied the memory phenotype of CD4 T cells using CD45RO and CD27 as markers for naïve (CD27<sup>pos</sup>CD45RO<sup>neg</sup>), "central memory-like" (CD27<sup>pos</sup>CD45RO<sup>pos</sup>) and effector memory (CD27<sup>neg</sup>CD45RO<sup>pos</sup>) CD4 T cells. Total memory CD4 T cells were defined as the sum of central memory (CD27<sup>pos</sup>CD45RO<sup>pos</sup>), effector memory (CD27<sup>neg</sup>CD45RO<sup>pos</sup>) and "terminally differentiated" (CD27<sup>neg</sup>CD45RO<sup>neg</sup>) CD4 T cells. *W. bancrofti* infection was associated with significantly increased frequencies of effector memory (CD27<sup>neg</sup>CD45RO<sup>pos</sup>) CD4 T cells, in univariable (mean 24.1% versus 20.0%, p = 0.0096) as well as multivariable analysis (Tables 3 and \$5). This was not observed for other analyzed helminth species. Importantly, the effector memory CD4 T cell frequencies were positively correlated with the activation status (Spearman's rho = 0.617, p<0.001, Fig 5).

Regulatory CD4 T cells, defined as CD25<sup>high</sup>FOXP3<sup>pos</sup> CD4 T cells were found in similar percentages in all study groups (Tables 3 and S4) with the exception of *T. trichiura*-infected individuals, who exhibited increased regulatory CD4 T cells compared to *T. trichiura*-uninfected individuals. Addressing the expression of HIV entry receptor CCR5, no difference was found regarding the percentage of CCR5 positive CD4 T cells, CCR5 positive regulatory CD4 T cells, or the mean fluorescence intensity of CCR5 on memory CD4 T cells between the different helminth infections (Tables 3, S6, S7 and S8).

#### Discussion

The importance of CD4 T cell activation for HIV susceptibility has been emphasized in several studies that focused on HIV exposed but seronegative individuals [18–20, 29–31]. Activated CD4 T cells are a major cellular reservoir for continuous HIV replication *in vivo*. Recently, a study embedded into a phase 2B HIV vaccine trial and conducted in a high-risk setting examined a variety of different immune cell populations and phenotypes in peripheral blood in relation to HIV acquisition. The study showed that of all studied cell subsets, the frequency of

	N	Mean	univariable			multivariable		
Covariate			Coef.	95% CI	p-value	Coef.	95% CI	p-value
Age								
(per year)	-	-	0,13	(0.08 to 0.18)	0.0000	0,14	(0.09 to 0.18)	0.0000
Sex								
female*	131	8,78	0,00	-	-	0,00	-	-
male	95	8,45	-0,56	(-1.71 to 0.58)	0.3332	0,15	(-0.92 to 1.21)	0.7887
Current fever								
no*	200	8,50	0,00	-	-	0,00	-	-
yes	20	10,39	1,99	(0.04 to 3.95)	0.0460	1,36	(-0.45 to 3.18)	0.1412
no data	6	7,39	-0,48	(-3.96 to 3.00)	0.7860	-0,13	(-3.35 to 3.09)	0.9359
W. bancrofti								
neg. *	193	8,31	0,00	-	-	0,00	-	-
pos.	33	10,59	1,83	(0.14 to 3.52)	0.0337	1,58	(0.08 to 3.09)	0.0392
Hookworm								
neg. *	147	8,83	0,00	-	-	0,00	-	-
pos.	79	8,29	-0,62	(-1.79 to 0.55)	0.3019	-0,74	(-1.83 to 0.35)	0.1827
A. lumbricoides								
neg. *	177	8,25	0,00	-	-	0,00	-	-
pos.	49	10,05	1,49	(0.11 to 2.86)	0.0337	2,12	(0.80 to 3.43)	0.0016
T. trichiura								
neg. *	189	8,34	0,00	-	-	0,00	-	-
pos.	37	10,15	1,15	(-0.53 to 2.83)	0.1807	1,82	(0.34 to 3.31)	0.0162
S. mansoni								
neg. *	145	8,64	0,00	-	-	0,00	-	-
pos.	81	8,63	-0,09	(-1.26 to 1.08)	0.8839	1,46	(0.28 to 2.65)	0.0155
S. haematobium								
neg. *	209	8,67	0,00	-	-	0,00	-	-
pos.	17	8,22	-0,48	(-2.60 to 1.64)	0.6573	0,71	(-1.26 to 2.67)	0.4818

Table 2. Association of helminth infections with percent of HLA-DR<sup>pos</sup> CD4 T cells. Uni- and multi-variable mixed-effects linear regression results, with random effect for residence in Kyela site, multivariable results adjusted for all variables shown.

N = number of observations; Mean = mean outcome; Coef. = coefficient; 95% CI = 95% confidence interval

\* reference stratum

https://doi.org/10.1371/journal.pntd.0007623.t002

activated HLA-DR<sup>pos</sup> CD4 T cells discriminated best between subjects who later HIV seroconverted and age-, ethnicity- and risk-matched controls who did not seroconvert, regardless of intervention arm [17].

To address potential mechanisms underlying the *W. bancrofti*-associated risk increase in HIV acquisition, which was recently described by our group [8], an evaluation of different markers for maturation, immune activation as well as CCR5 expression was therefore performed. We found significantly higher frequencies of HLA-DR<sup>pos</sup> and HLA-DR<sup>pos</sup>/CD38<sup>pos</sup> as well as "effector-like" memory (CD27<sup>neg</sup>CD45RO<sup>pos</sup>) CD4 T cells in participants infected with *W. bancrofti* compared to individuals with other helminth infections or without helminth infection.

Univariable analysis results were supported by multivariable analyses, in which potential confounders were addressed. In our previous publications about the EMINI cohort, we had described different age and gender distributions for the detected helminth species [22–25]. We therefore included age, gender, and additionally fever (during the last 24 hours as proxy for an

				Univariable*			Multivariable**		
Outcor	ne	N	Mean	Coef.	95% CI	p-value	Coef.	95% CI	p-value
% HLA	-DR <sup>pos</sup> cells of a	all CD4 T ce	lls						
	W. bancrofti								
	neg. *	193	8,31	0,00	-	-	0,00	-	-
	pos.	33	10,59	1,83	(0.14 to 3.52)	0.033	1,58	(0.08 to 3.09)	0.039
% HLA	-DR <sup>pos</sup> CD38 <sup>pos</sup>	cells of all C	CD4 T cells						
	W. bancrofti								
	neg. *	189	2,82	0,00	-	-	0,00	-	-
	pos.	31	3,69	0,86	(0.21 to 1.52)	0.009	0,84	(0.19 to 1.50)	0.011
% HLA	-DR <sup>pos</sup> cells of a	all CD8 T ce	lls						
	W. bancrofti								
	neg. *	192	20,8	0,00	-	-	0,00	-	-
	pos.	33	29,5	1,69	(-3.18 to 6.56)	0.496	0,47	(-3.96 to 4.89)	0.836
% HLA	-DR <sup>pos</sup> CD38 <sup>pos</sup>	cells of all C	D8 T cells						
	W. bancrofti								
	neg. *	189	7,69	0,00	-	-	0,00	-	-
	pos.	32	13,42	3,49	(0.94 to 6.04)	0.007	3,45	(0.91 to 5.98)	0.007
% CD2	7 <sup>neg</sup> CD45RO <sup>pos</sup>	cells of all C	CD4 T cells						
	W. bancrofti								
	neg. *	189	20,04	0,00	-	-	0,00	-	-
	pos.	31	24,10	4,06	(0.99 to 7.13)	0.009	3,16	(0.17 to 6.15)	0.038
% CD2	5 <sup>high</sup> FOXP3 <sup>pos</sup>	cells of all C	D4 T cells						
	W. bancrofti								
	neg. *	178	2,258	0,00	-	-	0,00	-	-
	pos.	30	2,453	0,13	(-0.30 to 0.57)	0.549	0,03	(-0.40 to 0.45)	0.898
% CCR	5 <sup>pos</sup> cells of all	CD4 T cells							
	W. bancrofti								
	neg. *	170	24,06	0,00	-	-	0,00	-	-
	pos.	30	25,15	1,09	(-2.29 to 4.47)	0.528	1,45	(-2.03 to 4.93)	0.414
% CCR	5 <sup>pos</sup> cells of reg	ulatory CD4	T cells						
	W. bancrofti								
	neg. *	170	54,47	0,00	-	-	0,00	-	-
	pos.	30	55,80	1,34	(-3.99 to 6.66)	0.622	2,55	(-2.89 to 8.00)	0.358
mean f	luorescence inte	ensity of CC	R5 on memory	CD4 T cells					
	W. bancrofti								
	neg. *	185	799	0,00	-	-	0,00	-	-
	pos.	31	699	-100,43	(-244.95 to 44.10)	0.173	-65,07	(-214.22 to 84.09)	0.392

Table 3. Associations of filariasis infection with different markers of immune activation. Separate uni- and multi-variable mixed-effects linear regression models for each marker with random effect for residence at Kyela site. Differences in number of observations are due to missing outcome data.

N = number of observations; Mean = mean outcome; Coef. = coefficient; 95% CI = 95% confidence interval

\* unadjusted, but with random effect for Kyela site

\*\* with random effect for Kyela site and adjusted for age, sex, current fever, and for hookworm, *A. lumbricoides*, *T. trichiura*, *S. mansoni & haematobium* infection. Full models showing all co-variates are contained in the supplement.

https://doi.org/10.1371/journal.pntd.0007623.t003

acute illness) into the model. In addition, we included the infection status of other helminths to adjust for potential confounding, since we had previously described a positive association of *Trichuris trichiura* and *Ascaris lumbricoides* with the expression of markers for systemic T cell activation [21]. The association between increased frequencies of activated HLA-DR<sup>pos</sup> and





HLA-DR<sup>pos</sup>/CD38<sup>pos</sup> and effector memory (CD27<sup>neg</sup>CD45RO<sup>pos</sup>) CD4 T cells in peripheral blood and *W. bancrofti* infection was independent of other host and environmental factors, suggesting a causal link between these two factors.

The participants studied here had been selected from the large general population cohort (EMINI) based on their helminth infection status. These participants were selected in 9 different study sites with distinct geographical conditions. Participants with *W. bancrofti* and *T. trichiura* infection were almost exclusively found in Kyela site close to Lake Nyasa, whereas hookworm, *A. lumbricoides* and *Schistosoma* infections were found in several study sites. We therefore included residence in Kyela site as a random effect into our multivariable model. The positive association of *W. bancrofti* with HLA-DR<sup>pos</sup> and HLA-DR<sup>pos</sup>/CD38<sup>pos</sup> as well as "effector-like" memory (CD27<sup>neg</sup>CD45RO<sup>pos</sup>) CD4 T cells was confirmed in the mixed linear regression models with area of residence as a random effect. In addition we measured increased percentage of HLA-DR<sup>pos</sup>/CD38<sup>pos</sup> CD8 T cells in *W. bancrofti*-infected individuals, however, there is no evidence that CD8 T cell may contribute to increased HIV susceptibility.

Our study has some limitations that we tried to address during the analysis. Because our cohort did not include *W. bancrofti* mono-infected individuals, we compared not only to helminth negative individuals, but also to individuals with other helminth infections. The participants in the study group with "other helminths" were infected with similar species in an almost identical composition compared to the *W. bancrofti* group, providing an optimal "background" to study the effect of an additional helminth species (Figs <u>3A</u> and <u>4A</u>). In addition, we showed that polyparasitism had no influence on immune activation (Figs <u>3B</u> and <u>4B</u>). This approach seems appropriate, as we might have missed more parasites, like *Onchocerca* 

*volvulus* and *Mansonella perstans*, *two other tissue nematodes*. Increasing information about the modulation of the immune response to other pathogens in co-infected individuals is available [32–36]. However, the study area is not known to be endemic for *O. volvulus* and we did not measure *M. perstans*.

In a previous publication [21], we described differences between helminth species regarding their association with T cell activation. With adjusting for other helminths in our multivariate analysis, we clearly showed that the described CD4 T cell activation in *W. bancrofti*- (but also in *A. lumbricoides and T. trichiura)*-infected individuals was independent of concomitant infection with other helminths. *W. bancrofti* infection was diagnosed by retrospective testing of bio-banked samples using an ELISA specific for circulating filarial antigen in plasma (Trop-Bio Og4C3 ELISA, Townsville, Australia), which was an identical diagnostic approach to what was used in our previous study that showed significantly increased HIV acquisition risk in *W. bancrofti* infection in more detail, such as investigating the presence of microfilariae (the offspring of the adult worms, responsible for *W. bancrofti* transmission from humans to mosquitos), which requires the nocturnal collection of blood. Unfortunately, we do not have information about microfilariae for our study participants.

Because the clinical presentation of W. bancrofti infection shows such variety, the immunological background of these features has been studied extensively: Hyporesponsiveness was reported from in vitro studies of PBMC cultures of W. bancrofti-infected individuals [37, 38]. Treatment of the filarial parasites resulted in recovery of antigen-specific responses supporting the direct causal relationship [39]. It was discovered that the hyporesponsiveness, which was described initially, was typical for the filarial infected but asymptomatic individuals, whereas in lymphedema patients Th1 and Th17 responses mediated the pathology [12]. However, the number of participants with pathology in our cohort was too small to describe differences among sub groups. Infected individuals, as measured by the presence of circulating filarial antigen, who do not have microfilariae in the night blood, display a distinct immunological profile, characterized by stronger filarial-specific interleukin-5, IL-10 and TNF-alpha responses, compared to microfilaremic individuals [40]. Regulatory T cells, which are responsible for tolerance to self-antigens, show stronger activity in microfilaremic, compared to amicrofilaremic individuals [41-44]. In this context, it is possible that also HLA-DR<sup>pos</sup> and HLA-DR<sup>pos</sup>/CD38<sup>pos</sup> CD4 T cell frequencies, and also expression of HIV entry receptors, may differ in microfilaremic, amicrofilaremic or symptomatic filarial infected patients. Further studies with a better characterization of participants are needed.

An increase of systemically activated CD4 T cells is likely to support early HIV dissemination because they are cellular targets for HIV replication [45, 46]. However, it is unknown whether increased activation of circulating T cells in peripheral blood is also linked to increased activation of CD4 T cells at the mucosal site of viral transmission. In our study, the frequency of HLA-DR<sup>pos</sup> CD4 T cells strongly correlates with the frequency of effector memory CD4 T cells and both subsets are significantly increased *W. bancrofti* infected subjects. We have previously shown in the same cohort that both CD27 negative effector memory CD4 T cells and HLA-DR+ memory CD4 T cells express significantly higher levels of the HIV coreceptor CCR5 compared to CD27+ central memory or naive CD4 T cells [21], which may facilitate better HIV entry into these specific cell subsets. These effector memory CD4 T cells may migrate into tissues [47], such as the mucosal surfaces of the reproductive tract. Thus, future studies need to study HIV co-receptor expression and T cell activation in the reproductive tract in relation to *W. bancrofti* infection.

**In summary,** our data clearly show that *W. bancrofti* infection is linked to systemic CD4 T cell activation, as determined by HLA-DR expression, providing a possible mechanism for the

increased HIV-susceptibility observed in *W. bancrofti*-infected individuals [8]. Whether treatment of *W. bancrofti* could decrease immune activation and whether this in turn could reduce HIV transmission in areas where lymphatic filariasis is prevalent needs to be examined in future studies.

## **Supporting information**

S1 Fig. Gating strategy for Lymphocytes, CD3<sup>pos</sup> T cells, CD4<sup>pos</sup> T cells, memory and naïve, CD4<sup>pos</sup> T cells, HLADR<sup>pos</sup>CD38<sup>pos</sup> CD4 T cells, and percentage of CCR5 positive CD3 <sup>pos</sup> CD4 <sup>pos</sup> T lymphocytes. (DOCX)

**S1 Table.** Association of various factors with percent of HLA-DR<sup>pos</sup>CD38<sup>pos</sup> cells of all CD4 T cells (N = 220; Mean = 2.95). Uni- and multi-variable mixed-effects linear regression results, with random effect for residence in Kyela site, multivariable models additionally adjusted for age, gender and fever during last 24 hours and different helminth infections. (DOCX)

**S2 Table.** Association of various factors with percent of HLA-DR<sup>pos</sup> cells of all CD8 T cells (N = 225; Mean = 22.1). Uni- and multi-variable mixed-effects linear regression results, with random effect for residence in Kyela site, multivariable models additionally adjusted for age, gender and fever during last 24 hours and different helminth infections. (DOCX)

S3 Table. Association of various factors with percent of HLA-DR<sup>pos</sup>CD38<sup>pos</sup> cells of all CD8 T cells (N = 221; Mean = 8.52). Uni- and multi-variable mixed-effects linear regression results, with random effect for residence in Kyela site, multivariable models additionally adjusted for age, gender and fever during last 24 hours and different helminth infections. (DOCX)

**S4 Table.** Association of various factors with percent of "effector memory" CD27<sup>neg</sup>C-D45RO<sup>pos</sup> cells of all CD4 T cells (N = 220; Mean = 20.62). Uni- and multi-variable mixedeffects linear regression results, with random effect for residence in Kyela site, multivariable models additionally adjusted for age, gender and fever during last 24 hours and different helminth infections.

(DOCX)

**S5 Table.** Association of various factors with percent of CD25<sup>high</sup>FOXP3<sup>pos</sup> cells of all CD4 T cells (N = 208; Mean = 2.287). Uni- and multi-variable mixed-effects linear regression results, with random effect for residence in Kyela site, multivariable models additionally adjusted for age, gender and fever during last 24 hours and different helminth infections. (DOCX)

**S6 Table.** Association of various factors with percent of CCR5<sup>pos</sup> cells of all CD4 T cells (N = 200; Mean = 24.22). Uni- and multi-variable mixed-effects linear regression results, with random effect for residence in Kyela site, multivariable models additionally adjusted for age, gender and fever during last 24 hours and different helminth infections. (DOCX)

S7 Table. Association of various factors with percentage of CCR5<sup>pos</sup> of all regulatory CD4 T cells (N = 200; Mean = 54.67). Uni- and multi-variable mixed-effects linear regression results, with random effect for residence in Kyela site, multivariable models additionally adjusted for age, gender and fever during last 24 hours and different helminth infections. (DOCX)

**S8 Table.** Association of various factors with mean fluorescence intensity of CCR5 on memory CD4 T cells (N = 216; Mean = 785). Uni- and multi-variable mixed-effects linear regression results, with random effect for residence in Kyela site, multivariable models additionally adjusted for age, gender and fever during last 24 hours and different helminth infections.

(DOCX)

#### **Author Contributions**

Conceptualization: Mkunde Chachage, Elmar Saathoff, Christof Geldmacher.

Data curation: Inge Kroidl, Petra Clowes, Elmar Saathoff, Christof Geldmacher.

Formal analysis: Inge Kroidl, Jonathan Mnkai, Myrna Berninghoff, Elmar Saathoff, Christof Geldmacher.

Funding acquisition: Michael Hoelscher, Elmar Saathoff, Christof Geldmacher.

Investigation: Christof Geldmacher.

**Methodology:** Inge Kroidl, Mkunde Chachage, Jonathan Mnkai, Myrna Berninghoff, Jaco J. Verweij.

Project administration: Anthony Nsojo, Petra Clowes.

**Supervision:** Inge Kroidl, Mkunde Chachage, Anthony Nsojo, Lucas Maganga, Nyanda E. Ntinginya, Leonard Maboko, Petra Clowes, Michael Hoelscher, Christof Geldmacher.

Validation: Mkunde Chachage.

Visualization: Inge Kroidl, Mkunde Chachage, Christof Geldmacher.

Writing - original draft: Inge Kroidl.

Writing – review & editing: Inge Kroidl, Mkunde Chachage, Petra Clowes, Michael Hoelscher, Elmar Saathoff, Christof Geldmacher.

#### References

- 1. UNAIDS. Global report: UNAIDS report on the global AIDS epidemic 2016. Geneva: Joint United Nations Programme on HIV/AIDS (UNAIDS). http://www.unaids.org/en/resources/fact-sheet (accesed on 03.01.2017).
- Bentwich Z, Kalinkovich A, Weisman Z. Immune activation is a dominant factor in the pathogenesis of African AIDS. Immunology today. 1995; 16(4):187–91. PMID: 7734046.
- Bentwich Z, Kalinkovich A, Weisman Z, Borkow G, Beyers N, Beyers AD. Can eradication of helminthic infections change the face of AIDS and tuberculosis? Immunology today. 1999; 20(11):485–7. PMID: 10529774.
- Bentwich Z, Kalinkovich A, Weisman Z, Grossman Z. Immune activation in the context of HIV infection. Clinical and experimental immunology. 1998; 111(1):1–2. <u>https://doi.org/10.1046/j.1365-2249.1998.</u> 00483.x PMID: 9472654; PubMed Central PMCID: PMC1904872.
- Kalinkovich A, Maayan S, Weisman Z, Harpaz N, Bentwich Z. Immune activation, a co-factor for HIV transmission in Thailand? Lancet. 1994; 343(8911):1506–7. https://doi.org/10.1016/s0140-6736(94) 92618-2 PMID: 7911209.
- Hotez PJ, Bundy DAP, Beegle K, Brooker S, Drake L, de Silva N, et al. Helminth Infections: Soil-transmitted Helminth Infections and Schistosomiasis. In: Jamison DT, Breman JG, Measham AR, Alleyne G, Claeson M, Evans DB, et al., editors. Disease Control Priorities in Developing Countries. 2nd ed. Washington (DC)2006.
- McSorley HJ, Maizels RM. Helminth infections and host immune regulation. Clin Microbiol Rev. 2012; 25(4):585–608. https://doi.org/10.1128/CMR.05040-11 PMID: 23034321; PubMed Central PMCID: PMC3485755.

- Kroidl I, Saathoff E, Maganga L, Makunde WH, Hoerauf A, Geldmacher C, et al. Effect of Wuchereria bancrofti infection on HIV incidence in southwest Tanzania: a prospective cohort study. Lancet. 2016; 388(10054):1912–20. https://doi.org/10.1016/S0140-6736(16)31252-1 PMID: 27495354.
- Babu S, Nutman TB. Proinflammatory cytokines dominate the early immune response to filarial parasites. J Immunol. 2003; 171(12):6723–32. https://doi.org/10.4049/jimmunol.171.12.6723 PMID: 14662876.
- O'Connor RA, Jenson JS, Osborne J, Devaney E. An enduring association? Microfilariae and immunosuppression [correction of immunosupression] in lymphatic filariasis. Trends Parasitol. 2003; 19 (12):565–70. PMID: 14642766.
- Anuradha R, George PJ, Kumaran P, Nutman TB, Babu S. Interleukin-10- and transforming growth factor beta-independent regulation of CD8(+) T cells expressing type 1 and type 2 cytokines in human lymphatic filariasis. Clin Vaccine Immunol. 2014; 21(12):1620–7. https://doi.org/10.1128/CVI.00598-14 PMID: 25253667; PubMed Central PMCID: PMC4248783.
- Babu S, Bhat SQ, Pavan Kumar N, Lipira AB, Kumar S, Karthik C, et al. Filarial lymphedema is characterized by antigen-specific Th1 and th17 proinflammatory responses and a lack of regulatory T cells. PLoS Negl Trop Dis. 2009; 3(4):e420. https://doi.org/10.1371/journal.pntd.0000420 PMID: 19381284; PubMed Central PMCID: PMC2666805.
- Babu S, Nutman TB. Immunopathogenesis of lymphatic filarial disease. Seminars in immunopathology. 2012; 34(6):847–61. https://doi.org/10.1007/s00281-012-0346-4 PMID: 23053393; PubMed Central PMCID: PMC3498535.
- Debrah AY, Mand S, Specht S, Marfo-Debrekyei Y, Batsa L, Pfarr K, et al. Doxycycline reduces plasma VEGF-C/sVEGFR-3 and improves pathology in lymphatic filariasis. PLoS Pathog. 2006; 2(9):e92. Epub 2006/10/19. 05-PLPA-RA-0274R4 [pii] https://doi.org/10.1371/journal.ppat.0020092 PMID: 17044733; PubMed Central PMCID: PMC1564427.
- Debrah AY, Mand S, Toliat MR, Marfo-Debrekyei Y, Batsa L, Nurnberg P, et al. Plasma vascular endothelial growth Factor-A (VEGF-A) and VEGF-A gene polymorphism are associated with hydrocele development in lymphatic filariasis. Am J Trop Med Hyg. 2007; 77(4):601–8. Epub 2007/11/06. 77/4/ 601 [pii]. PMID: 17978056.
- Lal RB, Kumaraswami V, Krishnan N, Nutman TB, Ottesen EA. Lymphocyte subpopulations in Bancroftian filariasis: activated (DR+) CD8+ T cells in patients with chronic lymphatic obstruction. Clinical and experimental immunology. 1989; 77(1):77–82. PMID: <u>2527654</u>; PubMed Central PMCID: PMC1541923.
- Nganou K. http://www.croiwebcasts.org/p/2017croi/croi33468. www.croiconferenceorg/sessions/ immune-correlates-hiv-acquisition.2017.
- Card CM, Rutherford WJ, Ramdahin S, Yao X, Kimani M, Wachihi C, et al. Reduced cellular susceptibility to in vitro HIV infection is associated with CD4+ T cell quiescence. PLoS One. 2012; 7(9):e45911. https://doi.org/10.1371/journal.pone.0045911 PMID: 23029309; PubMed Central PMCID: PMC3448692.
- Card CM, Ball TB, Fowke KR. Immune quiescence: a model of protection against HIV infection. Retrovirology. 2013; 10:141. https://doi.org/10.1186/1742-4690-10-141 PMID: 24257114; PubMed Central PMCID: PMC3874678.
- McLaren PJ, Ball TB, Wachihi C, Jaoko W, Kelvin DJ, Danesh A, et al. HIV-exposed seronegative commercial sex workers show a quiescent phenotype in the CD4+ T cell compartment and reduced expression of HIV-dependent host factors. J Infect Dis. 2010; 202 Suppl 3:S339–44. <u>https://doi.org/10.1086/655968 PMID: 20887221</u>.
- Chachage M, Podola L, Clowes P, Nsojo A, Bauer A, Mgaya O, et al. Helminth-associated systemic immune activation and HIV co-receptor expression: response to albendazole/praziquantel treatment. PLoS Negl Trop Dis. 2014; 8(3):e2755. https://doi.org/10.1371/journal.pntd.0002755 PMID: 24675895; PubMed Central PMCID: PMC3967945.
- 22. Kroidl I, Saathof E, Maganga L, Clowes P, Maboko L, Hoerauf A, et al. Prevalence of Lymphatic Filariasis and Treatment Effectiveness of Albendazole/ Ivermectin in Individuals with HIV Co-infection in Southwest-Tanzania. PLoS Negl Trop Dis. 2016; 10(4):e0004618. https://doi.org/10.1371/journal.pntd. 0004618 PMID: 27070786; PubMed Central PMCID: PMC4829227.
- Manz KM, Clowes P, Kroidl I, Kowuor DO, Geldmacher C, Ntinginya NE, et al. Trichuris trichiura infection and its relation to environmental factors in Mbeya region, Tanzania: A cross-sectional, population-based study. PLOS ONE. 2017; 12(4):e0175137. https://doi.org/10.1371/journal.pone.0175137 PMID: 28384306
- 24. Riess H, Clowes P, Kroidl I, Kowuor DO, Nsojo A, Mangu C, et al. Hookworm infection and environmental factors in mbeya region, Tanzania: a cross-sectional, population-based study. PLoS Negl Trop Dis.

2013; 7(9):e2408. https://doi.org/10.1371/journal.pntd.0002408 PMID: 24040430; PubMed Central PMCID: PMC3764225.

- Schule SA, Clowes P, Kroidl I, Kowuor DO, Nsojo A, Mangu C, et al. Ascaris lumbricoides infection and its relation to environmental factors in the Mbeya region of Tanzania, a cross-sectional, populationbased study. PLoS One. 2014; 9(3):e92032. https://doi.org/10.1371/journal.pone.0092032 PMID: 24643023; PubMed Central PMCID: PMC3958400.
- Mwangi MN, Roth JM, Smit MR, Trijsburg L, Mwangi AM, Demir AY, et al. Effect of Daily Antenatal Iron Supplementation on Plasmodium Infection in Kenyan Women: A Randomized Clinical Trial. Jama. 2015; 314(10):1009–20. https://doi.org/10.1001/jama.2015.9496 PMID: 26348751.
- Llewellyn S, Inpankaew T, Nery SV, Gray DJ, Verweij JJ, Clements AC, et al. Application of a Multiplex Quantitative PCR to Assess Prevalence and Intensity Of Intestinal Parasite Infections in a Controlled Clinical Trial. PLoS Negl Trop Dis. 2016; 10(1):e0004380. https://doi.org/10.1371/journal.pntd.0004380 PMID: 26820626; PubMed Central PMCID: PMC4731196.
- Pilotte N, Papaiakovou M, Grant JR, Bierwert LA, Llewellyn S, McCarthy JS, et al. Improved PCR-Based Detection of Soil Transmitted Helminth Infections Using a Next-Generation Sequencing Approach to Assay Design. PLoS Negl Trop Dis. 2016; 10(3):e0004578. https://doi.org/10.1371/ journal.pntd.0004578 PMID: 27027771; PubMed Central PMCID: PMC4814118.
- Koning FA, Otto SA, Hazenberg MD, Dekker L, Prins M, Miedema F, et al. Low-level CD4+ T cell activation is associated with low susceptibility to HIV-1 infection. J Immunol. 2005; 175(9):6117–22. <a href="https://doi.org/10.4049/jimmunol.175.9.6117">https://doi.org/10.4049/jimmunol.175.9.6117</a> PMID: 16237108.
- Begaud E, Chartier L, Marechal V, Ipero J, Leal J, Versmisse P, et al. Reduced CD4 T cell activation and in vitro susceptibility to HIV-1 infection in exposed uninfected Central Africans. Retrovirology. 2006; 3:35. https://doi.org/10.1186/1742-4690-3-35 PMID: 16792805; PubMed Central PMCID: PMC1524799.
- Camara M, Dieye TN, Seydi M, Diallo AA, Fall M, Diaw PA, et al. Low-level CD4+ T cell activation in HIV-exposed seronegative subjects: influence of gender and condom use. J Infect Dis. 2010; 201 (6):835–42. https://doi.org/10.1086/651000 PMID: 20136413.
- Babu S, Bhat SQ, Kumar NP, Jayantasri S, Rukmani S, Kumaran P, et al. Human type 1 and 17 responses in latent tuberculosis are modulated by coincident filarial infection through cytotoxic T lymphocyte antigen-4 and programmed death-1. J Infect Dis. 2009; 200(2):288–98. <u>https://doi.org/10.</u> 1086/599797 PMID: 19505258; PubMed Central PMCID: PMC2997351.
- 33. Ritter M, Ndongmo WPC, Njouendou AJ, Nghochuzie NN, Nchang LC, Tayong DB, et al. Mansonella perstans microfilaremic individuals are characterized by enhanced type 2 helper T and regulatory T and B cell subsets and dampened systemic innate and adaptive immune responses. PLoS Negl Trop Dis. 2018; 12(1):e0006184. https://doi.org/10.1371/journal.pntd.0006184 PMID: 29324739; PubMed Central PMCID: PMC5783424.
- Hoerauf A. Mansonella perstans—the importance of an endosymbiont. N Engl J Med. 2009; 361 (15):1502–4. Epub 2009/10/09. 361/15/1502 [pii] https://doi.org/10.1056/NEJMe0905193 PMID: 19812409.
- Hoerauf A, Satoguina J, Saeftel M, Specht S. Immunomodulation by filarial nematodes. Parasite Immunol. 2005; 27(10–11):417–29. Epub 2005/09/24. PIM792 [pii] https://doi.org/10.1111/j.1365-3024.2005.00792.x PMID: 16179035.
- Freedman DO, Lujan-Trangay A, Steel C, Gonzalez-Peralta C, Nutman TB. Immunoregulation in onchocerciasis. Functional and phenotypic abnormalities of lymphocyte subsets and changes with therapy. J Clin Invest. 1991; 88(1):231–8. <u>https://doi.org/10.1172/JCl115282</u> PMID: <u>1829096</u>; PubMed Central PMCID: PMC296024.
- King CL, Kumaraswami V, Poindexter RW, Kumari S, Jayaraman K, Alling DW, et al. Immunologic tolerance in lymphatic filariasis. Diminished parasite-specific T and B lymphocyte precursor frequency in the microfilaremic state. J Clin Invest. 1992; 89(5):1403–10. <u>https://doi.org/10.1172/JCI115729</u> PMID: 1569183; PubMed Central PMCID: PMC443009.
- Nutman TB, Kumaraswami V, Ottesen EA. Parasite-specific anergy in human filariasis. Insights after analysis of parasite antigen-driven lymphokine production. J Clin Invest. 1987; 79(5):1516–23. <u>https:// doi.org/10.1172/JCI112982 PMID: 3553242</u>; PubMed Central PMCID: PMC424428.
- Sartono E, Kruize YC, Kurniawan A, van der Meide PH, Partono F, Maizels RM, et al. Elevated cellular immune responses and interferon-gamma release after long-term diethylcarbamazine treatment of patients with human lymphatic filariasis. J Infect Dis. 1995; 171(6):1683–7. https://doi.org/10.1093/ infdis/171.6.1683 PMID: 7769319.
- 40. Arndts K, Deininger S, Specht S, Klarmann U, Mand S, Adjobimey T, et al. Elevated adaptive immune responses are associated with latent infections of Wuchereria bancrofti. PLoS Negl Trop Dis. 2012; 6

(4):e1611. https://doi.org/10.1371/journal.pntd.0001611 PMID: 22509424; PubMed Central PMCID: PMC3317915.

- Wammes LJ, Hamid F, Wiria AE, Wibowo H, Sartono E, Maizels RM, et al. Regulatory T cells in human lymphatic filariasis: stronger functional activity in microfilaremics. PLoS Negl Trop Dis. 2012; 6(5): e1655. https://doi.org/10.1371/journal.pntd.0001655 PMID: <u>22666510</u>; PubMed Central PMCID: PMC3362610.
- Metenou S, Nutman TB. Regulatory T cell subsets in filarial infection and their function. Front Immunol. 2013; 4:305. https://doi.org/10.3389/fimmu.2013.00305 PMID: 24137161; PubMed Central PMCID: PMC3786323.
- Ludwig-Portugall I, Layland LE. TLRs, Treg, and B Cells, an Interplay of Regulation during Helminth Infection. Front Immunol. 2012; 3:8. https://doi.org/10.3389/fimmu.2012.00008 PMID: 22566894; PubMed Central PMCID: PMC3342019.
- Metenou S, Dembele B, Konate S, Dolo H, Coulibaly SY, Coulibaly YI, et al. At homeostasis filarial infections have expanded adaptive T regulatory but not classical Th2 cells. J Immunol. 2010; 184 (9):5375–82. https://doi.org/10.4049/jimmunol.0904067 PMID: 20357251; PubMed Central PMCID: PMC3407820.
- Biancotto A, Iglehart SJ, Vanpouille C, Condack CE, Lisco A, Ruecker E, et al. HIV-1 induced activation of CD4+ T cells creates new targets for HIV-1 infection in human lymphoid tissue ex vivo. Blood. 2008; 111(2):699–704. https://doi.org/10.1182/blood-2007-05-088435 PMID: 17909079; PubMed Central PMCID: PMC2200839.
- Stevenson M, Stanwick TL, Dempsey MP, Lamonica CA. HIV-1 replication is controlled at the level of T cell activation and proviral integration. EMBO J. 1990; 9(5):1551–60. PMID: <u>2184033</u>; PubMed Central PMCID: PMC551849.
- Groot F, van Capel TM, Schuitemaker J, Berkhout B, de Jong EC. Differential susceptibility of naive, central memory and effector memory T cells to dendritic cell-mediated HIV-1 transmission. Retrovirology. 2006; 3:52. <u>https://doi.org/10.1186/1742-4690-3-52</u> PMID: <u>16916447</u>; PubMed Central PMCID: PMC1562438.